

# **Earth's Future**

# <mark>,</mark>

# **RESEARCH ARTICLE**

10.1029/2018EF001130

#### **Key Points:**

- Without agronomic adaptation, the dominant effect of temperature increase is to shorten growing periods and to reduce yields and production
- Adaptation via cultivars that maintain current growing periods under warming can compensate global production losses up to 2 K
- Irrigation would act as intensification rather than true adaptation, as it hardly affects the sensitivity of crop yields to warming

#### **Supporting Information:**

- Supporting Information S1
- Figure S1
- Figure S2
- Figure S3
- Figure S4
- Figure S5
- Figure S6
- Figure S7
- Figure S8
- Figure S9
- Figure S10
- Figure S11
- Figure S12
- Figure S13
- Figure S14
- Figure S15Figure S16
- Figure S16
- Figure S17

#### Correspondence to: S. Minoli.

sara.minoli@pik-potsdam.de

#### **Citation**:

Minoli, S., Müller, C., Elliott, J., Ruane, A. C., Jägermeyr, J., Zabel, F., et al. (2019). Global response patterns of major rainfed crops to adaptation by maintaining current growing periods and irrigation. *Earth's Future*, 7. https://doi.org/10.1029/ 2018EF001130

Received 7 DEC 2018 Accepted 2 SEP 2019 Accepted article online 30 SEP 2019

#### **Author Contributions**

**Conceptualization:** Sara Minoli, Christoph Müller, Joshua Elliott, Alex C. Ruane, Jonas Jägermeyr, Thomas A. M. Pugh

#### ©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Global Response Patterns of Major Rainfed Crops to Adaptation by Maintaining Current Growing Periods and Irrigation

Sara Minoli<sup>1,2,3</sup>, Christoph Müller<sup>1</sup>, Joshua Elliott<sup>4,5</sup>, Alex C. Ruane<sup>6</sup>, Jonas Jägermeyr<sup>1,4,6</sup>, Florian Zabel<sup>7</sup>, Marie Dury<sup>8</sup>, Christian Folberth<sup>9</sup>, Louis François<sup>8</sup>, Tobias Hank<sup>7</sup>, Ingrid Jacquemin<sup>8</sup>, Wenfeng Liu<sup>10,11</sup>, Stefan Olin<sup>12</sup>, and Thomas A. M. Pugh<sup>13,14</sup>

<sup>1</sup>Climate Resilience, Potsdam Institute for Climate Impact Research (PIK), Member of the Leibniz Association, 14412,Potsdam, Germany, <sup>2</sup>Integrative Research Institute on Transformations of Human Environment Systems (IRI THESys), Humboldt-Universität zu Berlin, Berlin, Germany, <sup>3</sup>Albrecht Daniel Thaer-Institut für Agrar- und Gartenbauwissenschaften, Humboldt-Universität zu Berlin, Berlin, Germany, <sup>4</sup>Department of Computer Science, University of Chicago, Chicago, IL, USA, <sup>5</sup>Center for Robust Decision-making on Climate and Energy Policy (RDCEP), University of Chicago, Chicago, IL, USA, <sup>6</sup>NASA Goddard Institute for Space Studies, New York, NY, USA, <sup>7</sup>Department of Geography, Ludwig-Maximilians-Universität München (LMU Munich), Munich, Germany, <sup>8</sup>Unité de Modélisation du Climat et des Cycles Biogéochimiques, UR SPHERES, Université de Liège, Liège, Belgium, <sup>9</sup>International Institute for Applied Systems Analysis, Ecosystem Services and Management Program, Laxenburg, Austria, <sup>10</sup>Eawag, Swiss Federal Institute of Aquatic Science and Technology, Duebendorf, Switzerland, <sup>11</sup>Laboratoire des Sciences du Climat et de L'Environnement LSCE/IPSL, CEA-CNRS-UVSQ, Université Paris-Saday, Gif-sur-Yvette, France, <sup>12</sup>Department of Physical Geography and Ecosystem Science, Lund University, Lund, Sweden, <sup>13</sup>School of Geography, Earth and Environmental Sciences, University of Birmingham, Birmingham, UK, <sup>14</sup>Birmingham Institute of Forest Research, University of Birmingham, Birmingham, UK

**Abstract** Increasing temperature trends are expected to impact yields of major field crops by affecting various plant processes, such as phenology, growth, and evapotranspiration. However, future projections typically do not consider the effects of agronomic adaptation in farming practices. We use an ensemble of seven Global Gridded Crop Models to quantify the impacts and adaptation potential of field crops under increasing temperature up to 6 K, accounting for model uncertainty. We find that without adaptation, the dominant effect of temperature increase is to shorten the growing period and to reduce grain yields and production. We then test the potential of two agronomic measures to combat warming-induced yield reduction: (i) use of cultivars with adjusted phenology to regain the reference growing period duration and (ii) conversion of rainfed systems to irrigated ones in order to alleviate the negative temperature effects that are mediated by crop evapotranspiration. We find that cultivar adaptation can fully compensate global production losses up to 2 K of temperature increase, with larger potentials in continental and temperate regions. Irrigation could also compensate production losses, but its potential is highest in arid regions, where irrigation expansion would be constrained by water scarcity. Moreover, we discuss that irrigation is not a true adaptation measure but rather an intensification strategy, as it equally increases production under any temperature level. In the tropics, even when introducing both adapted cultivars and irrigation, crop production declines already at moderate warming, making adaptation particularly challenging in these areas.

**Plain Language Summary** Global warming affects yields of grain crops, which are at the base of human diets. We use crop models to quantify its impacts on global crop production and to assess how adaptation could compensate for the adverse effects. We find that up to 2 K of increased temperature production can be maintained at the current level by using new cultivars, selected to maintain current growing period length under warming. Irrigation, as another management strategy, is shown to have the potential to increase yields in dry regions if water is available. However, models do not indicate that irrigation reduces the crops' sensitivity to warming. We find large differences in the yield response to warming and adaptation across climatic regions. While continental and temperate regions may benefit from higher temperatures but also show sizable adaptation potentials, tropical and arid regions show largest temperature impacts and smaller adaptation potentials. After all, these two crop management



Methodology: Sara Minoli, Christoph Müller, Joshua Elliott, Alex C. Ruane, Jonas Jägermeyr, Florian Zabel, Thomas A. M. Pugh Writing - Original Draft: Sara Minoli Formal Analysis: Sara Minoli Resources: Christoph Müller, Joshua Elliott, Florian Zabel, Marie Dury, Christian Folberth, Louis François, Tobias Hank, Ingrid Jacquemin, Wenfeng Liu, Stefan Olin, Thomas A. M. Pugh

Writing - review & editing: Sara Minoli, Christoph Müller, Joshua Elliott, Alex C. Ruane, Jonas Jägermeyr, Florian Zabel, Marie Dury, Christian Folberth, Louis François, Tobias Hank, Ingrid Jacquemin, Wenfeng Liu, Stefan Olin, Thomas A. M. Pugh options appear effective to balance the effects of moderate warming but cannot fully compensate impacts above 2 K of warming.

# 1. Introduction

Productivity of current cropping systems can be severely affected by changes in climatic and weather variables (Challinor et al., 2014; Rosenzweig et al., 2014). Increasing temperature trends have already negatively impacted productivity of agricultural crops over the last decades (Lobell et al., 2011). Multiple methodologies consistently estimate that warming of 1 K causes between 3.1% and 7.4% decline in actual yields of major cereal crops, if no adaptation measures are undertaken (Challinor et al., 2014; Liu et al., 2016; Zhao et al., 2017). Future projections indicate that large portions of current global harvested area will continue experiencing declines in the attainable yields, even under the assumption that management and technology could be transferred between regions, to areas where adaptation to climate change is most needed (Pugh et al., 2016).

Crop yield is a result of several physiological plant processes, many of which are mediated by the ambient temperature, as plants can only partially regulate their own temperature internally (Parent et al., 2010). Experimental evidence has shown that temperature increases up to a certain threshold level are associated with both accelerated rates of crop phenological development (the progress through the life cycle stages of the plant; Hatfield, 2016; Parent & Tardieu, 2012) and growth metabolism (e.g., photosynthesis and respiration; Atkin & Tjoelker, 2003; Von Caemmerer, 2000). If higher metabolic rates enhance primary productivity (biomass) per unit of time, faster phenology also leads to shorter crop growing period durations (time from sowing to maturity), which are often associated with shorter grain-filling periods and thus lower crop yields (Egli, 2011; Hatfield et al., 2011). High temperatures can reduce the net photosynthetic rate, because gross photosynthesis has a lower optimum than mitochondrial respiration (Yamori et al., 2014). High temperatures also reduce the carboxilation rate of Rubisco, increasing photorespiration in C<sub>3</sub> species (Ainsworth & Ort, 2010). Extreme temperatures can also permanently damage plant tissues and reduce yields. Grain crops are especially sensitive during the reproductive phase (Hatfield et al., 2018; Porter & Gawith, 1999), undergoing floret sterility and disruption of the pollination process leading to lower grain numbers (Farooq et al., 2011; Hatfield, 2016) and a slower grain filling rate (Rezaei et al., 2015). Although the increase in air temperature alone is not a sufficient condition for increasing the evaporative demand, temperature is among the key drivers of evapotranspiration rates (Donohue et al., 2010). Under nonlimiting water conditions, crop yields are enhanced by high rates of evapotranspiration, due to the coupled exchange of water and  $CO_2$ , increased by higher stomatal conductance. Vice versa, under limited water availability, if the increased evaporative demand cannot be fulfilled, stomatal conductance and hence yield are reduced (Passioura & Angus, 2010). Furthermore, higher evapotranspiration rates can deplete the soil water content faster, possibly leading to plant water stress (Bodner et al., 2015). The evapotranspiration cooling effect is the main process of canopy temperature self-regulation (Kimball, 2016). Most process-based crop models include temperature response functions on the major physiological rates, while only a few include heat-stress impact mechanisms and the canopy temperature regulation (Asseng et al., 2015; Atkin et al., 2005; Rezaei et al., 2015; Smith & Dukes, 2013; Wang et al., 2017; Webber et al., 2017). Moreover, the combined effects of different stresses, such as temperature and water, which often occur simultaneously, are still poorly understood and pose a challenge for current global crop modeling (Chenu et al., 2017).

Different agronomic management options have been proposed as adaptation strategies against temperature-induced yield losses. Most commonly, these include a shift of sowing dates, choice of cultivars with adjusted phenology, and irrigation (Challinor et al., 2014; Olesen et al., 2011; Parent et al., 2018; Ruiz-Ramos et al., 2018; Semenov et al., 2014; Tack et al., 2017). Sowing dates can be advanced or delayed to match the most favorable thermal conditions and to exploit a longer growing season (Olesen et al., 2012; Sacks et al., 2010; Waha et al., 2012). Cultivars with higher thermal-unit requirements (temperature accumulation above a certain base temperature to reach physiological maturity), different vernalization requirements (exposure to cold temperatures to induce flowering), or altered photoperiod sensitivity (development response to day length) can be used to counteract the shortening of the growing period due to temperature increase (Parent et al., 2018; Sacks & Kucharik, 2011). In turn, shorter maturing cultivars can help avoid terminal heat and water stress (Bodner et al., 2015; Mondal et al., 2013). Irrigation and other management strategies that increase soil moisture have the potential to compensate for the amplified

.....



Table					
Global Gridded Crop Models (GGCMs) Participating in the Study With Main References					
GGCM	M References				
CARA	AIB Dury et al. (2011) and Pirttioja et al. (2015)				
GEPI	C Liu et al. (2007) and Folberth et al. (2012)				
LPJ-G	UESS Lindeskog et al. (2013) and Olin et al. (2015)				
LPJm	L von Bloh et al. (2018)				
pDSS	AT Elliott et al. (2014) and Jones et al. (2003)				
PEPIC	W. Liu, Yang, Folberth, et al. (2016) and W. Liu, Yang, Liu, et al. (2016)				
PROM	MET Mauser and Bach (2009), Hank et al. (2015), and Mauser et al. (2015)				

evapotranspiration demand driven by temperature increase but also to alleviate heat stress and accelerated phenological progress, by cooling the canopy temperature (Siebert et al., 2017; Tack et al., 2017; Webber et al., 2017).

Although some studies assessed these adaptation strategies at local to regional scales (Burke & Emerick, 2016; Parent et al., 2018; Ruiz-Ramos et al., 2018; Semenov et al., 2014), their aggregated effects at the global scale remain an open question. Global projections of climate change impacts usually do not consider the adaptation potentials of agricultural system in response to climate change and might therefore overestimate impacts on crop yields and production.

Crop models allow for the conducting of virtual experiments to study the complex and interdependent biophysical effects of atmosphere and soil processes on crop growth and yield formation. As such, they are widely applied tools for the analysis of climate change impacts on agriculture and play a fundamental role in integrated assessment studies (Rosenzweig et al., 2018). Here we present the first spatially explicit global study of the adaptation potential of the major staple crops to local temperature increase in rainfed systems. We use results from the Global Gridded Crop Model (GGCM) Intercomparison (GGCMI) within the Agricultural Model Intercomparison and Improvement Project (Rosenzweig et al., 2013). The aims of the study are to assess the potential of adaptation in growing period selection and supplementary irrigation to avoid warming-induced reductions in crop yields. To this end, we study how uniform warming scenarios of 1, 2, 3, 4, and 6 K in each grid cell affect crop productivity and growing period duration. We then compare the impacts to a set of scenarios where hypothetical cultivars that maintain the reference growing period through adjusted phenology are introduced. As a second management measure, we study the effect of converting rainfed into irrigated systems. Both interventions are analyzed separately and jointly. Since crop model responses are uncertain and models often show complementary skills (Müller et al., 2017), we employ an ensemble of seven GGCMs to address associated uncertainties. Model simulations are run according to a harmonized protocol, in terms of both weather inputs and agronomic management settings. The analysis is focused on the five major staple crops: maize, winter wheat, spring wheat, rice, and soybean.

# 2. Materials and Methods

## 2.1. Simulation Protocol and Models

Seven GGCM frameworks (CARAIB, GEPIC, LPJ-GUESS, LPJmL, pDSSAT, PEPIC, and PROMET) contributed to this study (Table 1) and followed the GGCMI Phase 2 simulation protocol (refer also to section S1 in the supporting information). The GGCMs are a class of process-based crop simulation models that are built to perform simulations on crop productivity and other agroecological variables at the global scale (Rosenzweig et al., 2014). As other process-based models, they conceptually represent soil-plant-atmosphere biophysical processes, their interactions, and feedbacks. The processes are quantitatively expressed through mathematical functions, and their effects are integrated over time steps, so that the temporal dynamic of, for example, crop growth is explicitly simulated. Different crop species are represented through specific sets of processes, parameters, or functions, which are typically derived on experimental bases (Jones et al., 2017; Muller & Martre, 2019). In the GGCMs, inputs and outputs are grid based and have both spatial (longitude, latitude) and temporal (days, years) dimensions. Inputs include atmospheric  $CO_2$ , climate, soil, and agronomic management data. They estimate simultaneously multiple crop-specific variables, including yields, biomass production, maturity dates, and cumulative evapotranspiration.

All simulations were run at 0.5° spatial resolution and for 31 years of the historical climate (1980-2010); models with a daily time step used AgMERRA climate data (Ruane et al., 2015), while those with subdaily temporal resolution used ERA-Interim (Dee et al., 2011). We assume that the use of two different climate products does not affect the analysis much, as both data sets are observation based and were treated with the same perturbation approach (see below). Current cropland patterns were selected in model postprocessing. The grid cell- and crop-specific area was obtained from MIRCA2000 (Portmann et al., 2010) data set at 0.5° resolution. The experiment design consisted of separate simulations for five crops (maize, rice, soybean, spring wheat, and winter wheat), under one baseline temperature scenario (T0) and five levels of globally uniform temperature increases (T1, T2, T3, T4, and T6). Moreover, four management settings were simulated, which we call control management setting (T-sensitive growing period & Rainfed) and three adaptive management setting (Fixed growing period & Rainfed, T-sensitive growing period & Irrigation, and Fixed growing period & Irrigation). To target specific adaptation strategies, it is necessary to isolate the effect of individual climatic factors, as there is uncertainty in, for example, the temperature sensitivity to increased CO<sub>2</sub> and in future correlations between precipitation and temperature patterns (Carter et al., 2016; Schleussner et al., 2018; Zhao et al., 2017). In this study we aimed at isolating the impact of adaptation of crops and agricultural management to temperature increase. Therefore, the atmospheric CO<sub>2</sub> mixing ratio was kept constant at 360 ppmv in all simulation years and scenarios (see section 4). To verify whether our conclusions are independent from the  $CO_2$  mixing ratio, we repeated the experiment also at 660 ppmv. For the comparison, we rely on a smaller set of GGCM, because GEPIC and PEPIC did not provide the full CO<sub>2</sub> offsets simulations (Figure S15). Similarly, precipitation and other climate drivers were unchanged across scenarios. The five artificial warming scenarios were created by perturbing input daily air temperature by five respective offsets (+1, +2, +3, +4, and +6 K).

The model ensemble was harmonized for three key management practices: (1) the growing period, (2) the water supply (rainfed or fully irrigated), and (3) the nitrogen-fertilizer application rate (assumed to be 200 kg  $N \cdot ha^{-1} \cdot year^{-1}$  uniformly for each crop and cropping season, applied in two doses: 50% at planting and 50% on a crop- and grid-specific day; see protocol in section S1). The growing period harmonization followed the protocol of GGCMI Phase 1 (Elliott et al., 2015), based on observed growing period data (Portmann et al., 2010; Sacks et al., 2010), gap filled with rule-based (Waha et al., 2012) cropping calendars. Modelers were asked to calibrate the phenology, so that the average (over the 31-year simulation period) growing periods matched the provided crop- and grid-specific sowing and maturity dates. Sowing dates were kept constant at the historical observations. Maturity dates were estimated from observed harvest dates by subtracting crop-specific maturity-to-harvest times (21, 7, 21, 7, and 7 days for maize, rice, soybean, spring wheat, and winter wheat, respectively) from the latter (Elliott et al., 2015). The procedure for the calibration to observed growing periods was individually chosen by each modeling team, which could freely determine phenological parameters such as cardinal temperatures, growing degree days, vernalization, and/or photoperiod requirements, as well as set these as grid-specific or global values. The obtained parametrization was assumed to describe the available historical crop cultivar pool (see details in Tables 2 and S1).

In addition to simulations assuming control management, three adaptive management scenarios were simulated for each temperature level. Under the fixed growing period setting, we assumed the use of different hypothetical cultivars with adapted phenological traits, which maintain the reference growing period under each warming level. This represents a measure to counteract the higher-temperature effect on the phenological development rate, by which the time between sowing and maturity is generally shortened. All GGCMs use thermal time as the main driver of phenological progress (Table 2); therefore, higher air temperature offsets are expected to affect the growing period durations. The fixed growing period implementation was simulated by adjusting the crop phenological parameters, so that the average (over the 31-year simulation period) length of the growing period (in days) was the same (as closely as possible) under all TO-T6 scenarios. Therefore, modelers were asked to implement individual solutions to maintain the 1980-2010 mean growing period extent (e.g., precalculating changes in thermal time requirements based on fixed temperature shifts or adjusting by iteration). For models that separate phenology into multiple stages (e.g., sowing-to-anthesis and anthesis-to-maturity), modelers were asked to scale parameters of each stage equally, so that the timing of intermediate stages such as anthesis stayed approximately the same. Under the *irrigation* setting, we assumed the supply of unlimited irrigation water to the crops. Irrigation is studied as an adaptation measure to temperature increase because there are interactions between crop temperature and water supply. Higher air temperatures can increase the rates of evapotranspiration and soil water depletion, while



GGCMs Participating in the Study With Main Features of Their Phenological Module					
GGCM	Temperature response function	Phenological drivers	Perceived Temperature	Phenological phases	
CARAIB	Lin., Tmin	T(GDD), W	Tair	1 (S-M)	
GEPIC	Lin., Tmin, Topt	T(GDD), T(V), DL	Tair	1 (S-M)	
LPJ-GUESS	Lin., Tmin, Topt*, Tmax*	T(GDD), T(V)	Tair	2 (S-A-M)	
LPJmL	Lin., Tmin	T(GDD), T(V)	Tair	1 (S-M)	
pDSSAT	Lin., Tmin, Topt	T(GDD), T(V), DL, W	Tair	Crop specific	
PEPIC	Lin., Tmin, Topt	T(GDD), T(V), DL	Tair	1 (S-M)	
PROMET	Curv., Tmin, Topt, Tmax	T(DVR), T(V), DL, W	Tleaf	100 (BBCH)	

Table 2

Note. More details are reported in section S2 and Table S1. Temperature response function for phenology (Wang et al., 2017): Lin = linear; Curv. = curvilinear; Tmin = minimum cardinal temperature; Topt = optimum cardinal temperature; Tmax = maximum cardinal temperature; asterisk (\*) denotes for spring wheat and winter wheat only. Phenological drivers: T(GDD) = temperature (growing degree days); T(DVR) = temperature (development rate); T(V) = temperature (vernalization); DL = daylength; W = water; N = nitrogen. Perceived temperature: Temperature perceived by the crop, driving phenological and metabolic processes (Tair = air temperature; Tleaf = leaf temperature). Phenological phases: S = sowing; A = anthesis; M = maturity; BBCH = full BBCH. GGCMs = Global Gridded Crop Models.

> soil water deficits can reduce evapotranspiration and thus increase the canopy temperature, consequently affecting the phenological development and growth rates. Although not all these effects are included in the GGCMs, each of them represents water-to-temperature interaction in some ways. All the ensemble's GGCMs simulate evapotranspiration by formulas that require temperature as an input variable; thus, temperature increases are expected to modify the crop water demand. Moreover, some GGCMs represent the feedback between water stress and phenology. Particularly, two models include water deficit as directly affecting phenological progress (Table 2). In pDSSAT, water deficit may delay the onset of reproductive growth, while it accelerates the grain filling phase (Jones et al., 2003), while in CARAIB, the water deficit may delay germination. Only the PROMET model includes the indirect effect of soil water status on phenology, through the explicit simulation of the leaves temperature (Table 2). Water deficit results in increased leaf temperature that usually accelerates the phenological progress. Irrigation was assumed to be unconstrained by surface water availability. Irrigation was implemented to refill soil water content to field capacity as soon as it fell below a threshold of 90% of field capacity (Elliott et al., 2015). We also tested the combination of fixed growing period and irrigation.

#### 2.2. Model Output Processing

Models reported yearly dry-matter yields (Mg/ha), sowing dates (day of year), and maturity dates (days from planting) for the period 1980-2010, separately for maize, rice, soybean, spring wheat, and winter wheat. Yield failures were reported as 0 Mg/ha, while nonsimulated grids were reported as NA values. Maturity dates of yield failure years were set to NA.

We computed the long-term averages (1981-2009) for yield, sowing date, and maturity date for the period 1981-2009 for each model, temperature offset scenario, and management setting. The first and last years of the simulation time series were excluded to avoid reporting issues relating to completeness of growing periods (Elliott et al., 2015).

Under the reference temperature scenario (T0), the growing periods were assumed to be the same in all management settings, and thus, yields were assumed to be the same as well for the rainfed and irrigation settings. For efficiency reasons, outputs for T5 were not simulated but derived by linear interpolation between T4 and T6 for each GGCM, crop, and grid cell, independently for each of the management setting (categorical variables). Some individual simulations were not available for all models (Figure S1), and we gap filled missing simulations by linear interpolation of neighboring scenarios (with an exception for rice and soybean for the LPJ-GUESS model that were not simulated at all and for winter wheat for the PEPIC model that was excluded from the analysis due to unreliable simulations of the growing periods). The CARAIB model had only sowing dates harmonized, while the model was not calibrated to match observed harvest dates (see section S2.1 for details), but the simulation with fixed growing periods are unaffected.

All GGCM output data are made publicly available at zenodo.org repository. The DOI references are provided in Table 3. For data processing, we used R (R Core Team, 2018) and R-packages for handling netcdf4



Table 3 DOL References for Accessing the Data Lload in This Study								
Dor References for Accessing the Data Osea in This Study								
GGCM	Maize	Soybean	Rice	Winter wheat	Spring wheat			
CARAIB	2582522	2582508	2582504	2582516	2582499			
GEPIC	2582247	2582258	2582251	2582260	2582263			
LPJ-GUESS	2581625	_	—	2581638	2581640			
LPJmL	2581356	2581498	2581436	2581565	2581606			
pDSSAT	2582111	2582147	2582127	2582163	2582178			
PEPIC	2582341	2582433	2582343	2582439	2582455			
PROMET	2582467	2582488	2582479	2582490	2582492			
	DOI References J GGCM CARAIB GEPIC LPJ-GUESS LPJmL pDSSAT PEPIC PROMET	DOI References for Accessing I           GGCM         Maize           CARAIB         2582522           GEPIC         2582247           LPJ-GUESS         2581625           LPJmL         2581356           pDSSAT         2582341           PEPIC         2582341           PROMET         2582467	DOI References for Accessing the Data Used it           GGCM         Maize         Soybean           CARAIB         2582522         2582508           GEPIC         2582247         2582258           LPJ-GUESS         2581356         —           LPJmL         2581356         2581498           pDSSAT         2582341         2582433           PROMET         2582467         2582488	DOI References for Accessing the Data Used in This Study           GGCM         Maize         Soybean         Rice           CARAIB         2582522         2582508         2582504           GEPIC         2582247         2582258         2582251           LPJ-GUESS         2581625         —         —           LPJmL         2581356         2581498         2581436           pDSSAT         2582311         2582433         2582343           PROMET         2582467         2582488         2582479	DOI References for Accessing the Data Used in This Study           GGCM         Maize         Soybean         Rice         Winter wheat           CARAIB         2582522         2582508         2582504         2582516           GEPIC         2582247         2582258         2582251         2582260           LPJ-GUESS         2581625         —         —         2581638           LPJmL         2581356         2582147         2582147         2582163           pDSSAT         2582311         2582433         2582433         2582439           PROMET         2582467         2582488         2582479         2582490			

*Note*. All GGCMs' output data, separated by crop and model, can be found at https://doi.org/10. 5281/zenodo/XX, where XX is the value reported in the table. GGCM = Global Gridded Crop Models.

(Pierce, 2015), performing computation (Dowle & Srinivasan, 2017; Wickham, 2011), and plotting results (Wickham, 2009).

## 2.3. Metrics

All GGCMs included in this study simulate crop phenology as a function of temperature (thermal-unit sum and vernalization). We quantified the average impact of globally uniform temperature increase on growing periods and yields across all GGCMs and cropland grid cells by fitting linear regression models for each individual crop (Figures 2a and 2b). To understand whether there is a direct relationship between responses of growing periods and yields to temperature increase, we analyzed the joint distribution of their changes from the reference scenario (T0). We categorized the possible responses into four classes, defined by the sign of change of the two variables, and illustrated their frequency of occurrence within each class and climatic regions (Figure 2c).

To quantify the impact of temperature increase on global production of all crops, we estimated the production change (%) under warming scenarios as compared to the reference temperature scenario (T0). Since here we considered production of crops of relevance for human nutrition, we transform yields from metric tons to their calorie content, as this is the most common metric for quantifying globally available food (Willett et al., 2019). The grid-based global calories production under the management system *m* and temperature offset *n* ( $P_{m,n}$ , equation (1)) was obtained as the sum of production across all crops (*c*) and grid cells (g). Within the grid cell *j*, the yield of crop *i* was multiplied by its calorie content and by its area in that grid.

$$P_{m,n} = \sum_{j=1}^{g} \sum_{i=1}^{c} area_{j,i} \cdot yield_{j,i} \cdot calorie_{i}.$$
(1)

The calorie content values were derived from the FAO (2001) food balance sheet handbook, which reports food composition in terms of weight "as purchased"; therefore, model output yields were converted from dry to fresh matter as from Wirsenius (2000) to obtain the calorie yield per crop and unit of area (Table 4).

#### Table 4

Crop-Specific Parameters Used for Converting the Crop Yield Model Outputs From Metric Tons of Dry Matter (Mg/ha) to Their Calorie Content (Gcal/ha)

Crop	Grain dry matter	Calorie content	
	(% as-purchased)	(Gcal/Mg)	
Maize	88	3.560	
Soybean	91	3.350	
Spring wheat	88	3.340	
Winter wheat	88	3.340	
Rice	87	2.800	

*Note.* The grain dry matter (% of as-purchased grain weight) is the yield conversion factor between as-purchased-grain weight and dry-grain weight. The calorie content are the calories per metric ton of as-purchased grain weight.





**Figure 1.** Diagram of the Adaptation Index (*AI*) computation (modified from Lobell, 2014). The plot shows the yield (Mg/ha) as a function of increasing temperature between the reference climate and the offset temperature  $T_n$  (ranging from T1 to T6). The black and the red lines represent the Yield ~ Temperature response with *control* and *adaptive* crop management, respectively. The red dashed lines delimit the space where *adaptive* management only partially compensate yield losses (0 < AI < 100). *AI* is computed as in equation (2) where *a* is the yield impact of the temperature increase from  $T_0$  to  $T_n$  on yield under *control* management, *b* is the effect that the adaptive management would have under the reference conditions ( $T_0$ ), and *c* is the effect of the *adaptive* management under the offset temperature  $T_n$ .

To determine whether the simulated adaptation measures are effective, we computed the *Adaptation Index* (*AI*, equation (2); modified from Lobell, 2014) for each grid cell, temperature offset, and *adaptive management* setting as

$$AI = 100 \cdot (c - b)/|a|, \ if \ a < 0, \tag{2}$$

where *a* is the impact of temperature increase on yield under *control management*, *b* is the effect that the *adaptive management* would have under the Reference Scenario T0, and *c* is the effect of the *adaptive management* under increased temperature scenarios (Figure 1). Note that by definition under T0, the *fixed growing period* is equal to *T*-sensitive growing period, and therefore, *b* is zero for this *adaptive management* scenario. Values of *AI* are computed only if *a* is negative; otherwise, temperature increase is considered beneficial. *AI* ranges between  $-\infty$  and  $+\infty$ , with  $AI \ge 100$  indicating full compensation or overcompensation of losses (full adaptation), 0 < AI < 100 indicating partial compensation of losses (partial adaptation), and AI < 0 indicating no compensation of losses, meaning either an amplification of damages or that the *adaptive management* can be increasing production, without being impact reducing (intensification), and therefore not a true adaptation measure (Lobell, 2014). *AI* was computed for each single GGCM, and we then computed the median ensemble and uncertainty (range) across GGCMs.

### 3. Results

# 3.1. Crop Phenology Response to Temperature Offsets Without Adaptation

At the global aggregation level, in absence of adaptation measures, the growing period length approximates a negative linear response to increasing temperature from 1 to 6 K (Figure 2). The slope of this relationship (days of growing period lost per Kelvin of warming) is similar across the five simulated crops, ranging from 5.4 (maize) to 3.8 days/K (spring wheat). The spread of growing period length across all GGCMs and all cropland globally does not change fundamentally at higher temperature offsets (whiskers in Figure 2), yet it somewhat increases for soybean and decreases for winter wheat. The general response is similar across the GGCMs as most models simulate shortening growing periods with higher temperatures (Figure S4). Yet some models show smaller sensitivity of the growing period change across crops (PEPIC and PROMET)



(a) Temperature effect on growing period

(b) Temperature effect on yield



count

Figure 2. Effect of increasing temperature on (a) crop phenology (growing period duration in days from planting) and (b) yield (Mg/ha), separated by the five simulated main staple crops, without any adaptation measure. Each box represents the 25th, 50th, and 75th percentile of all grid cells values of all Global Gridded Crop Models for a specific temperature offset. The whiskers extends from the hinge to the smallest and largest values, no further than 1.5 interquartile range. The red line is the linear regression across simulated values and temperature offsets, and its slope is indicated in each plot. The heat map in (c) displays the relationship between yield changes (Mg/ha) and growing period changes (number of days) for all crops and all temperature offsets, separated by the Koeppen-Geiger climate regions. Changes are calculated as the absolute differences between the reference and the offset scenario. Plot in (a) and (b) includes all cultivated grid cells at all temperature offsets between T1 and T6; note that T5 is not simulated but linearly interpolated from T4 and T6. T5 is not included in (c). Hexagons are colored according to their frequency count.

and, in one case, an opposite sign of change (rice for PROMET). In PROMET, phenology is implemented to slow down at high temperatures above a crop-specific optimum (Table 2), so that warming can also lead to growing periods longer than in the reference temperature scenario. This is the case for LPJ-GUESS (spring and winter wheat) as well, although parametrized with higher optimum thresholds (Table 2), which could be the reason for the nonoccurrence of such increase. In some GGCMs (LPJmL and PROMET), for winter wheat, vernalization requirements are not satisfied as quickly under warming, so that the phenological development decelerates. Spatial patterns of growing period length at different temperature levels show that its shortening (days) is especially pronounced in cold-temperature limited regions for maize and soybean (Figure S8a), because under warming, more days reach temperatures above the crop-specific base temperatures ( $T_{min}$ ).

#### 3.2. Impact of Temperature Offsets on Crop Yield

The crop yield response follows similar patterns as the growing period, with an almost linear decline with temperature at the global aggregation level. Yield declines range between 0.39 Mg·ha<sup>-1</sup>·K<sup>-1</sup> for rice and 0.13 Mg·ha<sup>-1</sup>·K<sup>-1</sup> for winter wheat (Figure 2b). As opposed to the growing period length, the spread of the GGCMs declines with higher temperature offsets. For rice, soybean, and winter wheat, we also observe narrowing interquartile ranges (Figure 2b) with higher temperature offsets. This reflects a stronger reduction of high yields than of low yields. Indeed, for some regions and models, yields are already null under the baseline climate and cannot decrease further.

Changes in growing period and changes in crop yields do not follow a one-to-one relationship, but in most cases (~69% of all cultivated grid cells, across crops and temperature offsets), shorter growing periods are associated with declining yields (Figures 2c and S8): 69%, 61%, 68%, 69%, and 11% for tropical, arid, temperate, continental, and polar areas, respectively. There are rare cases where decreasing growing periods are associated with increasing yields or where longer growing periods are associated with declining yields. The former can be explained either by the beneficial effect of overall higher growing season temperatures that stimulate primary productivity, as for maize in high latitudes, or by the benefit of a shorter growing season that escapes water stress, if water conditions become more limiting under increased temperatures. Similarly, longer growing periods can lead to yield decline if the growing periods extend into dry season. Longer growing periods with increasing crop yields are, however, basically nonexistent.

#### 3.3. Effectiveness of Adaptation Measures

Increasing temperatures decrease global production of all crops almost linearly, except for winter wheat, where decrease only starts at a warming of  $\sim 2$  K (red line in Figure 3a). Using different cultivars so that the original growing period would be maintained, the total global calorie production of all five crops can be stabilized up to  $\sim 2$  K and declines with further warming (*fixed growing period* setting; yellow lines in Figure 3a; all crops). Individual crops, however, show different temperature responses. Rice, soybean, and spring wheat show an almost linear decline with any warming. For maize, the *fixed growing period* setting leads to stable global production up to warming of  $\sim 3$  K. For winter wheat, warming of up to  $\sim 4$  K is projected to even increase global production and decreases only thereafter. At the global aggregation level, the agreement across GGCMs is reasonable, except for large uncertainty for spring wheat (yellow shaded area in Figure 3a). All models show benefits from the *fixed growing period* setting, compared to *control management*. For all crops except rice, there is always at least one GGCM that simulates increase global production with the fixed growing period setting up to >5 K.

Converting all currently rainfed to irrigated cropland (assuming unlimited water supply; irrigation setting) would increase global calorie production by  $\sim 20\%$  (blue line in Figure 3a). At such higher level, fully irrigated production would also decline with increasing temperatures at similar rates to rainfed production. Yet the intensification through irrigation could maintain current production levels up to  $\sim$ 4 K at global level across crops. Similar patterns are displayed by all crops, while irrigated winter wheat may be able to maintain current production levels even up to >6 K. In combination, the fixed growing period and irrigation measures support global calorie production increases up to 4 K temperature offests (green line in Figure 3a) and could maintain current levels across all tested warming levels. For maize, the combined implementation of both measures leads to continuous increases in production across all tested warming levels. Global rice production seems to be the least sensitive to the two adaptation options at any temperature offset. Across the different GGCMs, the effects of irrigation on currently rainfed cropland are generally more uncertain than the effects of warming on rainfed crop production. The size of the response of spring wheat production to introducing irrigation to all rainfed areas is highly uncertain across GGCMs, as two models (LPJmL and pDSSAT) project roughly doubled spring wheat production under irrigation, whereas the other models project increases of only 25%. Still, as for the other crops, the uniform decline of irrigated spring wheat production is found by all GGCMs.

The level of warming up to which global production of all crops can increase differs across regions and management scenarios (Figure 3b). Particularly, in the tropics under the *fixed growing period*, global rainfed



# Earth's Future



**Figure 3.** Effect of increasing temperatures and four management settings on global calorie production. (a) Calories production change (percent change), shown as an aggregation for each crop. The lines and the shaded areas represent the median and the range of the Global Gridded Crop Models ensemble, respectively. For visualizing the full range of the spring wheat crop, refer to Figure S11. (b) Global patterns of temperature offsets up to which calories production of all crops increases under each *adaptive management* scenario, compared to the *control management* in the baseline temperature scenario (T0).

production is already declining at 1 K of warming, whereas in temperate and continental zones, it increases up to high levels of warming (also 6 K). *Irrigation* would maintain higher production levels up to 6 K in arid regions, where however irrigation expansion would be difficult due to the scarce water resources. In temperate and tropical regions where *irrigation* would likely be more feasible, this measure is effective until moderate warming only or not effective at all like in many tropical areas. Only the combination of *fixed growing period* and *irrigation* maintains crop production above the original level in largest part of the global land. Yet the tropical areas show production declines at 1 to 4 K of warming.

Increases in crop production are in some cases also a direct effect of temperature increase, even in absence of management changes, as in high-latitude regions (dark green in Figure 4a). Although the three management options show potentials to increase crop productivity globally, they are not necessarily true adaptation measures. Particularly, we find that despite the positive effect of *irrigation* in leverage yields, it does not (AI < 0) or only partially (0 < AI < 100) reduce the negative effect of warming (shortening of the growing period) but rather overcompensate them. The effectiveness of the *fixed growing period* differs across regions (Figure 4a). Figure 4a shows in detail the effectiveness of the *fixed growing period* adaptation for a warming level of 4 K; however, the patterns are very similar across all warming levels tested here (see Figure S13). In temperate and continental regions, there is larger potential for adaptation through the *fixed growing period* than in tropical and arid regions. The *fixed growing periods* measure has hardly any positive effect in arid regions but has the potential to maintain current production levels in continental regions (see Figure S12). In arid regions, the *fixed growing period* measure can lead to amplified damages and would thus be a form of "maladaptation" (orange color in Figures 4a, S13, and S14). The models here capture the typical interaction mechanism between plant phenology and water use. The selection of earlier-maturing cultivars in environments characterized by water shortage is a well-known strategy for avoiding water stress to crops (Bodner

Rice

Soybean

Spring wheat

Winter wheat







**Figure 4.** Adaptation potential of rainfed crops under a 4 K temperature offset in combination with *fixed growing period* adaptation scenario (a) and model agreement (b). (a) shows Global Gridded Crop Model ensemble medians for (i) negative temperature impact and negative adaptation effect (orange, a < 0 & AI < 0); (ii) negative temperature impact and negative adaptation effect (orange, a < 0 & AI < 0); (ii) negative temperature impact and positive adaptation effect, with full compensations (blue, a < 0 & 0 < AI < 100); (iii) negative temperature impact and positive adaptation effect, with full compensation (light green, a < 0 & AI > 100); (iv) positive temperature impacts (dark green, a > 0); and (v) neutral impacts (gray, a = 0). (b) shows the range of AI across Global Gridded Crop Models, only in the grid cells where temperature increase has negative impacts (a < 0). Values larger than 200% are constrained to 200% for better visualization. AI and a are computed as in equation (2).

et al., 2015). Under increasing temperatures, the atmospheric water demand increases as well, which cannot be fulfilled by actual evapotranspiration. Extending the growing period therefore worsens water stress, determining a maladaptation effect. There are substantial differences in the global patterns of adaptation effectiveness across the GGCMs (Figures 4b and S13), with a larger agreement on where *fixed growing period* has little adaptation effectiveness, but models often disagree (larger ensemble *AI* range) on the magnitudes of the adaptation effectiveness of the *fixed growing period* measure (Figure 4b).

In tropical regions, *irrigation* has little potential to intensify production, whereas it has substantial potential in arid regions with the highest levels of water stress (see Figure S12). However, availability to realize these potentials are not considered here.

Under higher  $CO_2$  mixing ratio of 660 ppmv, temperature impacts with the *control management* are slightly reduced compared to those under 360 ppmv. However, the findings on *fixed growing period* and *irrigation* adaptation potentials hold valid also when assuming higher  $CO_2$  mixing ratio (Figures S16–S18). For spring wheat only, the adaptation potential of the *fixed growing period* setting is larger under 660 ppmv than at 360 ppmv, so that global level production can be maintained across all temperature offsets.

# 4. Discussion

Using a large ensemble of GGCMs in a systematic warming experiment, we find that temperature increases lead to continuous reductions in global crop production without compensating adaptation measures, which is in line with previous findings (Challinor et al., 2014; Lobell et al., 2011; Liu et al., 2016; Rosenzweig et al., 2014; Zhao et al., 2017). Our results suggest that this decline is driven by a combination of accelerated phenology and thus shorter growing periods as well as by direct effects on plant growth. As such, selecting cultivars that maintain the original growing period under warming is a viable adaptation measure in most regions, as it reduces or fully compensates negative effects of warming on crop yields. This confirms recent findings in that historical warming already leads to the use of longer maturing cultivars, which in turn contributed to the increasing yield trend in the United States and Europe (Butler et al., 2018; Parent et al., 2018). The response, however, is variable across regions, crops, and GGCMs and thus subject to uncertainties.

In absence of detailed information on crop and cultivar parameters, process-based models applied at global scale have to make critical assumptions. Folberth et al. (2016) highlights how important the assumptions on management aspects are for simulating crop yields. Simulated adaptation potentials at global scale are necessarily affected by coarse model assumptions as well. Previous GGCMI ensemble studies showed that harmonization of management settings (Elliott et al., 2015) can have substantial effects on model performance (Müller et al., 2017); however, only a small set of settings can be harmonized, limited by the availability of global data sets (e.g., fertilizer and growing periods). Recently, Jägermeyr and Frieler (2018) highlighted that the correct timing of the growing season is particularly critical for realistic yield simulation. In this exercise, participating modelers were therefore asked to parametrize crop phenology so that current observed growing periods (Elliott et al., 2015) were reproduced by each model (by calculating required thermal units based on AgMERRA weather data; Ruane et al., 2015). In the simulations without fixed growing period adaptation (T-sensitive growing period setting), simulated growing periods were allowed to respond to warming, depending on the individual GGCMs' implementation of phenology (Table 2). In the fixed growing period adaptation setting, the crop phenology parameters were recalibrated for each warming level, so that the growing period length was roughly unaffected. However, no harmonization was requested for any other cultivar parameters or the functional form of the phenological response to temperature (Table 2). To this end, the ensemble of GGCMs used here reflects a broader variety of cultivars and management systems, which may explain part of the diverse modeled regional response to fixed growing period adaptation. Cardinal temperatures of phenological development (Table S1) are considered crop specific (Hatfield et al., 2011), with very little variability within species among genotypes and no acclimation to changes in temperature (Parent & Tardieu, 2012), therefore supporting the use by the GGCMs of crop-specific global parameters in the temperature response function for phenology. On the other hand, photosynthesis and enzyme activity acclimate to higher temperature (Parent & Tardieu, 2012) and cultivars differ in their sensitivity to heat stress. In particular, cultivars that are selected in hot climates are less sensitive to yield losses (Butler & Huybers, 2013), a feature that is not reflected in the GGCMs. Rezaei et al. (2018) suggest that the temperature response in phenology could be flawed by not accounting for changes in cultivar choice in the historic past. Also, Zhu et al. (2019) find that GGCMs often overestimate the response in growing period length compared to other yield-reducing effects of warming. This uncertainty in regional responses is in contrast to the robust finding of *fixed growing period* adaptation at the global aggregation, where models do not differ much. It remains unclear how the diverse regional responses need to be aggregated. Considering the complex interaction of initial cultivar parameterization for baseline yields and warming effects (Folberth et al., 2016), this requires further research and requires better information on existing management systems globally. In this study we assume the availability of the adapted cultivars to maintain the growing periods. However, it is not clear whether these would actually be available today or in the future, and how large would the effort be for breeding such new cultivars, especially under elevated temperatures (Challinor et al., 2016).

While irrigation comes with substantial potential to lift yields in water-limited regions, therefore buffering temperature-induced adverse effects, converting rainfed to irrigated crop production cannot be considered as a true adaptation measure but rather intensification (Lobell, 2014). This is because the beneficial effect of irrigation is similar under warming and under current conditions. This is surprising, because previous studies suggest that irrigation reduces the direct negative effects of warming on crop yields. Schauberger et al. (2017) find that irrigation buffers against damages from exposure to hot temperatures for maize in the United States and maize yield response to temperature is found to be highly leveraged by soil moisture status (Carter et al., 2016) and thus presumably irrigation in dryer areas. Observed yield declines during historical heat waves and droughts are predominantly attributed to rainfed systems (Jägermeyr & Frieler, 2018). The lack of reproducing this effect in this model ensemble may be due to several reasons. First, only one of the seven GGCMs (PROMET) accounts for the cooling effect of increased transpiration under irrigation by simulating canopy temperature, whereas all other models assume canopy temperature to be equal to air temperature. However, PROMET also shows the same pattern in the response to irrigation and warming as all the other GGCMs: intensification of production through converting rainfed to irrigated production but no benefit on the negative response to warming (see Figure S11). Second, if models tend to overestimate the growing period response to warming, as suggested by Rezaei et al. (2018) and Zhu et al. (2019), it may be that the shortening of the growing season overly dominates the yield response and direct effects of warming on plant growth are underrepresented in the current GGCMs. If canopy temperatures are not accounted for, irrigation cannot affect the simulated length of the growing period and will not show the underrepresented effects on crop growth. Nonetheless, this intensification could compensate for much of the warming-induced damage, at least in regions where irrigation water could be supplied.

We find that the challenge to maintain current productivity levels under warming is particularly large in the tropical and arid climate zones, given that the adaptation of *fixed growing period* has little potential to reduce the negative effects of warming on crop yields and that shifting rainfed to irrigated production has little potential in the tropics and will be severely hampered by water availability in most arid regions. Furthermore, large areas of the tropics are bound to experience climate conditions that have no analogs under current climate conditions (Pugh et al., 2016), so that breeding or designing cultivars for such conditions will be particularly challenging.

Generally, the GGCMI Phase 2 modeling experiment is an artificial setup with several implications for the interpretation of results (Supporting Information S1). First, we study the effects of warming in a uniform manner, that is, all days warmed by exactly the same offset, which is not representative for realistic climate change scenarios. Second, associated impacts from changes in precipitation under climate change are ignored, while direct impacts of elevated atmospheric CO<sub>2</sub> mixing ratios are tested in a rather simplistic manner. This can substantially enhance crop growth (Kimball, 2016) but also amplify high temperature damage to crops (Prasad et al., 2006). However, we find that our findings regarding adaptation potential to temperature increase remain valid under both CO2 mixing ratios. Third, we here consider only high-input systems with nitrogen fertilization levels of 200 kg N/ha and no other nutrient limitations (e.g., Phosphorus and Potassium). However, these simplifications seem justified, as we are aiming to understand how warming-induced damages to crop production can be compensated by an adaptation measure that counteracts the phenological acceleration, which is almost exclusively temperature driven. This sensitivity study helps to isolate effects, which are typically difficult to separate in realistic climate scenarios, in which the relationship of changes in temperature, precipitation, and atmospheric CO<sub>2</sub> mixing ratios are very model dependent (McSweeney & Jones, 2016). Nevertheless, results from the GGCMI CTWN-A experiment as analyzed here should not be misinterpreted as assessments of adaptation options under realistic climate change scenarios.

The adaptation measure to regain the warming-induced loss of growing period duration considered here is also a simplified theoretical case. As the maintaining of the original growing period is not always beneficial or somewhat shorter or longer growing periods could have even greater potential to adapt to warming, this only represents one specific case of a broader continuum. Also, adaptation in cultivar choice is likely linked to changes in sowing dates to best adapt cropping systems and exploit benefits of a longer growing season, as it is already observed with already contributes to recent yield trends (Butler et al., 2018; Parent et al., 2018). Flexible sowing dates could thus further increase the adaptation potentials shown here as we assume static sowing dates even under extreme temperature offsets. This is especially relevant in regions with temperature seasonality (Waha et al., 2012). Shifting sowing dates has nonetheless its complications, especially for crops with sensitivity to photoperiod and or vernalization, as it could in turn affect the length of the growing period, with importance consequences on the final yields (Abdulai et al., 2012; Hunt et al., 2019). Moreover, we analyze winter and spring wheat separately, therefore maintaining their crop areas static as though they were two different crops. This is a simplification, as these are rather two varieties of the same species, of which the spatial distribution depends on temperature and particularly on the existence of a winter season suitable for vernalization. A change in temperature level can then make it more advantageous to switch between winter and spring varieties, than trying to maintain the original growing season as an adaptation option. However, selecting a static growing season as a uniform adaptation measure again facilitates better interpretation of the results. Still, uncertainties are linked to the differences in interpreting the modeling protocol. The simulations conducted by CARAIB did not parametrize cultivar traits to reproduce the harmonization target for crop maturity (Elliott et al., 2015) but do keep their growing seasons constant under warming in the adaptation setup. Other models did follow the harmonization protocol, but growing seasons are not always closely reproduced (see Figure S2). This contributes to the uncertainty in the modeled response to adaptation.

# 5. Conclusions

Without agronomic adaptation measures, future warming as projected under climate change will negatively affect global crop production in absence of adaptation measures. Despite possibly compensating or amplifying effects from simultaneously changing precipitation and atmospheric CO<sub>2</sub> concentrations, it is important to understand temperature-driven plant physiological processes, in order to identify adaptation options to warming-induced yield reductions. By using a GGCM ensemble, we find that adaptation via new cultivars that would maintain current crop growing periods under warming is a viable option with substantial potential to fully compensate warming-induced yield reductions, especially in the temperate and continental climate zones. Even though growing period adaptation also shows positive effects in the tropics but hardly any in arid regions, these effects are insufficient to fully compensate warming-induced yield reduction even at low levels of warming. Tropical regions are also not very responsive to introducing irrigated production systems so that maintaining current crop productivity under warming is particularly challenging in the tropics. Here we have used the largest available data set of model simulations at global scale to present the most comprehensive approach for simulating temperature impacts on crop yields. That said, scarcity of global-scale management input data renders simulations at this scale inherently uncertain. Yet we find good agreement on the globally aggregated impacts and effects of adaptation across the GGCMs that implicitly represents a broader set of management systems by differing in the parametrization of management-related features. Future research will have to explore the potential for adaptation and intensification in temperate and continental climate zones to contribute to future food security and will have to identify ways how the double burden of strong climate change impacts and low adaptation potential in the tropical and arid climate zones can be alleviated.

# References

Abdulai, A., Kouressy, M., Vaksmann, M., Asch, F., Giese, M., & Holger, B. (2012). Latitude and date of sowing influences phenology of photoperiod-sensitive sorghums. *Journal of Agronomy and Crop Science*, 198(5), 340–348.

Ainsworth, E. A., & Ort, D. R. (2010). How do we improve crop production in a warming world? *Plant physiology*, 154(2), 526–530.

Asseng, S., Zhu, Y., Wang, E., & Zhang, W. (2015). Crop modeling for climate change impact and adaptation, (2nd ed.)., *Crop physiology* pp. 505–546). San Diego: Elsevier. http://www.sciencedirect.com/science/article/pii/B9780124171046000200

Atkin, O. K., Bruhn, D., Hurry, V. M., & Tjoelker, M. G. (2005). Evans review no. 2: The hot and the cold: Unravelling the variable response of plant respiration to temperature. *Functional Plant Biology*, 32(2), 87–105.

#### Acknowledgments

Data are made available at a central open-access repository (zenodo.org). DOIs are referenced in Table 3. S. M. and C. M. acknowledge financial support from the MACMIT project (01LN1317A) funded through BMBF (Germany). A. C. R. received support from the NASA Earth Sciences Division via the NASA GISS Climate Impacts Group (USA). J. J. received funding from the Open Philanthropy Project with partial support from the U. Chicago RDCEP center (NSF grant #SES-146364), F. Z. acknowledges financial support from the BioNex Project (031B0230B) funded by BMBF (Germany). W. L. was funded by the Early Postdoctoral Mobility Fellowship of the Swiss National Science Foundation (P2EZP2 175096), T. A. M. P. was supported by the European Commission's 7th Framework Programme, under Grant Agreement 603542 (LUC4C). This is paper number 40 of the Birmingham Institute of Forest Research.

Atkin, O. K., & Tjoelker, M. G. (2003). Thermal acclimation and the dynamic response of plant respiration to temperature. *Trends in plant science*, 8(7), 343–351.

Bodner, G., Nakhforoosh, A., & Kaul, H.-P. (2015). Management of crop water under drought: A review. Agronomy for Sustainable Development, 35(2), 401-442.

Burke, M., & Emerick, K. (2016). Adaptation to climate change: Evidence from US agriculture. American Economic Journal: Economic Policy, 8(3), 106–40.

Butler, E. E., & Huybers, P. (2013). Adaptation of US maize to temperature variations. Nature Climate Change, 3(1), 68-72.

Butler, E. E., Mueller, N. D., & Huybers, P. (2018). Peculiarly pleasant weather for US maize. Proceedings of the National Academy of Sciences, 115(47), 11.935–11.940.

Carter, E. K., Melkonian, J., Riha, S. J., & Shaw, S. B. (2016). Separating heat stress from moisture stress: Analyzing yield response to high temperature in irrigated maize. *Environmental Research Letters*, 11(9), 94012.

Challinor, A. J., Koehler, A.-K., Ramirez-Villegas, J., Whitfield, S., & Das, B. (2016). Current warming will reduce yields unless maize breeding and seed systems adapt immediately. *Nature Climate Change*, 6(10), 954.

Challinor, A. J., Watson, J., Lobell, D., Howden, S., Smith, D., & Chhetri, N. (2014). A meta-analysis of crop yield under climate change and adaptation. *Nature Climate Change*, 4(4), 287.

Chenu, K., Porter, J. R., Martre, P., Basso, B., Chapman, S. C., Ewert, F., et al. (2017). Contribution of crop models to adaptation in wheat. *Trends in Plant Science*, 22(6), 472–490.

Dee, D. P., Uppala, S., Simmons, A., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. Quarterly Journal of the Royal Meteorological Society, 137(656), 553–597.

Donohue, R. J., McVicar, T. R., & Roderick, M. L. (2010). Assessing the ability of potential evaporation formulations to capture the dynamics in evaporative demand within a changing climate. *Journal of Hydrology*, *386*(1-4), 186–197.

Dowle, M., & Srinivasan, A. (2017). data.table: Extension of 'data.frame'. https://CRAN.R-project.org/package=data.table, R package version 1.10.4-3.

Dury, M., Hambuckers, A., Warnant, P., Henrot, A., Favre, E, Ouberdous, M., & François, L. (2011). Responses of European forest ecosystems to 21(st) century climate: Assessing changes in interannual variability and fire intensity. *iForest: Biogeosciences and Forestry*, 4, 82–99.

Egli, D. B. (2011). Time and the productivity of agronomic crops and cropping systems. Agronomy journal, 103(3), 743–750.

Elliott, J., Kelly, D., Chryssanthacopoulos, J., Glotter, M., Jhunjhnuwala, K., Best, N., Wilde, M., & Foster, I. (2014). The parallel system for integrating impact models and sectors (pSIMS). *Environmental Modelling & Software*, 62, 509–516, 1364-8152. https://doi.org/10.1016/ j.envsoft.2014.04.008, http://www.sciencedirect.com/science/article/pii/S1364815214001121

Elliott, J., Müller, C, Deryng, D., Chryssanthacopoulos, J., Boote, K., Büchner, M, et al. (2015). The Global Gridded Crop Model Intercomparison: Data and modeling protocols for phase 1 (v1. 0). *Geoscientific Model Development*, 8(2), 261–277.

FAO (2001). Food balance sheets: A handbook. Food and Agriculture Organization, Rome, Italy, http://www.fao.org/docrep/003/x9892e/X9892e05.htm#P8217\_125315

Farooq, M., Bramley, H., Palta, J. A., & Siddique, K. H. (2011). Heat stress in wheat during reproductive and grain-filling phases. Critical Reviews in Plant Sciences, 30(6), 491–507.

Folberth, C., Elliott, J., Müller, C., Balkovic, J., Chryssanthacopoulos, J., Izaurralde, R. C., et al. (2016). Uncertainties in global crop model frameworks: Effects of cultivar distribution, crop management and soil handling on crop yield estimates. *Biogeosciences Discussions*, 2016, 1–30. https://www.biogeosciences-discuss.net/bg-2016-527/

Folberth, C., Gaiser, T., Abbaspour, K. C., Schulin, R., & Yang, H. (2012). Regionalization of a large-scale crop growth model for sub-Saharan Africa: Model setup, evaluation, and estimation of maize yields. *Agriculture, ecosystems & environment*, 151, 21–33.

Hank, T. B., Bach, H., & Mauser, W. (2015). Using a remote sensing-supported hydro-agroecological model for field-scale simulation of heterogeneous crop growth and yield: Application for wheat in central Europe. *Remote Sensing*, 7(4), 3934–3965.

Hatfield, J. L. (2016). Increased temperatures have dramatic effects on growth and grain yield of three maize hybrids. Agricultural & Environmental Letters, 1(1), 1–5. https://dl.sciencesocieties.org/publications/ael/abstracts/1/1/150006

Hatfield, J. L., Boote, K. J., Kimball, B., Ziska, L., Izaurralde, R. C., Ort, D., et al. (2011). Climate impacts on agriculture: Implications for crop production. Agronomy journal, 103(2), 351–370.

Hatfield, J., Wright-Morton, L., & Hall, B. (2018). Vulnerability of grain crops and croplands in the midwest to climatic variability and adaptation strategies. *Climatic Change*, 146(1-2), 263–275.

Hunt, J. R., Lilley, J. M., Trevaskis, B., Flohr, B. M., Peake, A., Fletcher, A., et al. (2019). Early sowing systems can boost australian wheat yields despite recent climate change. *Nature Climate Change*, 9(3), 244.

Jägermeyr, J., & Frieler, K. (2018). Spatial variations in crop growing seasons pivotal to reproduce global fluctuations in maize and wheat yields. *Science Advances*, 4(11), eaat4517.

Jones, J. W., Antle, J. M., Basso, B., Boote, K. J., Conant, R. T., Foster, I., et al. (2017). Brief history of agricultural systems modeling. Agricultural systems, 155, 240–254.

Jones, J. W., Hoogenboom, G., Porter, C. H., Boote, K. J., Batchelor, W. D., Hunt, L., et al. (2003). The DSSAT cropping system model. *European Journal of Agronomy*, 18(3-4), 235–265.

Kimball, B. A. (2016). Crop responses to elevated CO<sub>2</sub> and interactions with H<sub>2</sub>O, N, and temperature. *Current Opinion in Plant Biology*, 31, 36-43.

Lindeskog, M., Arneth, A., Bondeau, A., Waha, K., Seaquist, J., Olin, S., & Smith, B. (2013). Implications of accounting for land use in simulations of ecosystem carbon cycling in Africa. *Earth System Dynamics*, 4(2), 385–407.

Liu, B., Asseng, S., Müller, C., Ewert, F., Elliott, J., Lobell, D. B., et al. (2016). Similar estimates of temperature impacts on global wheat yield by three independent methods. *Nature Climate Change*, 6(12), 1130.

Liu, J., Williams, J. R., Zehnder, A. J. B., & Yang, H. (2007). GEPIC—Modelling wheat yield and crop water productivity with high resolution on a global scale. *Agricultural systems*, 94(2), 478–493.

Liu, W., Yang, H., Folberth, C., Wang, X., Luo, Q., & Schulin, R. (2016). Global investigation of impacts of PET methods on simulating crop-water relations for maize. Agricultural and forest meteorology, 221, 164–175.

Liu, W., Yang, H., Liu, J., Azevedo, L. B., Wang, X., Xu, Z., et al. (2016). Global assessment of nitrogen losses and trade-offs with yields from major crop cultivations. *Science of The Total Environment*, 572, 526–537.

Lobell, D. B. (2014). Climate change adaptation in crop production: Beware of illusions. Global Food Security, 3(2), 72-76.

 Lobell, D. B., Schlenker, W., & Costa-Roberts, J. (2011). Climate trends and global crop production since 1980. Science, 333(6042), 616–620.
 Müller, C., Elliott, J., Chryssanthacopoulos, J., Arneth, A., Balkovic, J., Ciais, P., et al. (2017). Global Gridded Crop Model evaluation: Benchmarking, skills, deficiencies and implications. Geoscientific Model Development Discussions, 10, 1403–1422.

- Mauser, W., & Bach, H. (2009). PROMET—Large scale distributed hydrological modelling to study the impact of climate change on the water flows of mountain watersheds. *Journal of Hydrology*, 376(3-4), 362–377.
- Mauser, W., Klepper, G., Zabel, F., Delzeit, R., Hank, T., Putzenlechner, B., & Calzadilla, A. (2015). Global biomass production potentials exceed expected future demand without the need for cropland expansion. *Nature communications*, *6*, 8946.
- McSweeney, C. F., & Jones, R. G. (2016). How representative is the spread of climate projections from the 5 CMIP5 GCMs used in ISI-MIP? *Climate Services*, 1, 24–29.
- Mondal, S., Singh, R., Crossa, J., Huerta-Espino, J., Sharma, I., Chatrath, R., et al. (2013). Earliness in wheat: A key to adaptation under terminal and continual high temperature stress in South Asia. *Field crops research*, 151, 19–26.
- Muller, B., & Martre, P. (2019). Plant and crop simulation models: Powerful tools to link physiology, genetics, and phenomics. UK: Oxford University Press.
- Olesen, J. E., Børgesen, C. D., Elsgaard, L., Palosuo, T., Rötter, R., Skjelvåg, A., et al. (2012). Changes in time of sowing, flowering and maturity of cereals in Europe under climate change. Food Additives & Contaminants: Part A, 29(10), 1527–1542.
- Olesen, J. E., Trnka, M., Kersebaum, K. C., Skjelvåg, A., Seguin, B., Peltonen-Sainio, P., et al. (2011). Impacts and adaptation of European crop production systems to climate change. *European Journal of Agronomy*, 34(2), 96–112.
- Olin, S., Schurgers, G., Lindeskog, M., Wårlind, D., Smith, B., Bodin, P., et al. (2015). Modelling the response of yields and tissue C: N to changes in atmospheric CO<sub>2</sub> and N management in the main wheat regions of western Europe. *Biogeosciences*, *12*(8), 2489–2515.
- Parent, B., Leclere, M., Lacube, S., Semenov, M. A., Welcker, C., Martre, P., & Tardieu, F. (2018). Maize yields over Europe may increase in spite of climate change, with an appropriate use of the genetic variability of flowering time. *Proceedings of the National Academy of Sciences*, 115(42), 10.642–10.647.
- Parent, B., & Tardieu, F. (2012). Temperature responses of developmental processes have not been affected by breeding in different ecological areas for 17 crop species. *New Phytologist*, 194(3), 760–774.
- Parent, B., Turc, O., Gibon, Y., Stitt, M., & Tardieu, F. (2010). Modelling temperature-compensated physiological rates, based on the co-ordination of responses to temperature of developmental processes. *Journal of Experimental Botany*, 61(8), 2057–2069.
- Passioura, J., & Angus, J. (2010). Improving productivity of crops in water-limited environments, *Advances in agronomy* (vol. 106, pp. 37–75). Amsterdam, The Netherlands: Elsevier.
- Pierce, D. (2015). ncdf4: Interface to unidata netcdf (version 4 or earlier) format data files [Computer software manual]. Retrieved from http://CRAN.R-project.org/package=ncdf4 (R package version 1.15).
- Pirttioja, N., Carter, T. R., Fronzek, S., Bindi, M., Hoffmann, H., Palosuo, T., et al. (2015). Temperature and precipitation effects on wheat yield across a European transect: A crop model ensemble analysis using impact response surfaces. *Climate Research*, 65, 87–105.
- Porter, J. R., & Gawith, M. (1999). Temperatures and the growth and development of wheat: A review. European journal of agronomy, 10(1), 23–36.
- Portmann, F. T., Siebert, S., & Döll, P. (2010). MIRCA2000—Global monthly irrigated and rainfed crop areas around the year 2000: A new high-resolution data set for agricultural and hydrological modeling. *Global Biogeochemical Cycles*, 24(1), GB1011.
- Prasad, P. V., Boote, K. J., & Allen Jr., L. H. (2006). Adverse high temperature effects on pollen viability, seed-set, seed yield and harvest index of grain-sorghum [Sorghum bicolor (L.) Moench] are more severe at elevated carbon dioxide due to higher tissue temperatures. *Agricultural and forest meteorology*, 139(3-4), 237–251.
- Pugh, T. A. M., Müller, C., Elliott, J., Deryng, D., Folberth, C., Olin, S., et al. (2016). Climate analogues suggest limited potential for intensification of production on current croplands under climate change. *Nature Communications*, 7, 12608.
- R Core Team (2018). R: A language and environment for statistical computing [Computer software manual]. Vienna, Austria. Retrieved from https://www.R-project.org/
- Rezaei, E. E., Siebert, S., Hüging, H., & Ewert, F. (2018). Climate change effect on wheat phenology depends on cultivar change. Scientific reports, 8(1), 4891.
- Rezaei, E. E., Webber, H., Gaiser, T., Naab, J., & Ewert, F. (2015). Heat stress in cereals: Mechanisms and modelling. *European Journal of Agronomy*, 64, 98–113.
- Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A. C., Müller, C., Arneth, A., et al. (2014). Assessing agricultural risks of climate change in the 21st century in a Global Gridded Crop Model Intercomparison. *Proceedings of the National Academy of Sciences*, 111(9), 3268–3273.
- Rosenzweig, C., Jones, J. W., Hatfield, J. L., Ruane, A. C., Boote, K. J., Thorburn, P., et al. (2013). The Agricultural Model Intercomparison and Improvement Project (AgMIP): Protocols and pilot studies. Agricultural and Forest Meteorology, 170, 166–182.
- Rosenzweig, C., Ruane, A. C., Antle, J., Elliott, J., Ashfaq, M., Chatta, A. A., et al. (2018). Coordinating AgMIP data and models across global and regional scales for 1.5° C and 2.0° C assessments. *Philosophical Transactions of the Royal Society A*, 376(2119), 20160455.
- Ruane, A. C., Goldberg, R., & Chryssanthacopoulos, J. (2015). Climate forcing datasets for agricultural modeling: Merged products for gap-filling and historical climate series estimation. *Agricultural and Forest Meteorology*, 200, 233–248.
- Ruiz-Ramos, M., Ferrise, R., Rodríguez, A, Lorite, I., Bindi, M., Carter, T. R., et al. (2018). Adaptation response surfaces for managing wheat under perturbed climate and CO<sub>2</sub> in a mediterranean environment. *Agricultural Systems*, 159, 260–274.
- Sacks, W. J., Deryng, D., Foley, J. A., & Ramankutty, N. (2010). Crop planting dates: An analysis of global patterns. *Global Ecology and Biogeography*, 19(5), 607–620.
- Sacks, W. J., & Kucharik, C. J. (2011). Crop management and phenology trends in the us corn belt: Impacts on yields, evapotranspiration and energy balance. Agricultural and Forest Meteorology, 151(7), 882–894.
- Schauberger, B., Archontoulis, S., Arneth, A., Balkovic, J., Ciais, P., Deryng, D., et al. (2017). Consistent negative response of US crops to high temperatures in observations and crop models. *Nature communications*, *8*, 13931.
- Schleussner, C.-F., Deryng, D., Müller, C., Elliott, J., Saeed, F., Folberth, C., et al. (2018). Crop productivity changes in 1.5° C and 2° C worlds under climate sensitivity uncertainty. *Environmental Research Letters*, *13*(6), 64007.
- Semenov, M., Stratonovitch, P., Alghabari, F., & Gooding, M. (2014). Adapting wheat in Europe for climate change. Journal of Cereal Science, 59(3), 245–256.
- Siebert, S., Webber, H., Zhao, G., & Ewert, F. (2017). Heat stress is overestimated in climate impact studies for irrigated agriculture. Environmental Research Letters, 12(5), 54023.
- Smith, N. G., & Dukes, J. S. (2013). Plant respiration and photosynthesis in global-scale models: Incorporating acclimation to temperature and CO<sub>2</sub>. Global Change Biology, 19(1), 45–63.
- Tack, J., Barkley, A., & Hendricks, N. (2017). Irrigation offsets wheat yield reductions from warming temperatures. *Environmental Research Letters*, 12(11), 114027.
- von Bloh, W., Schaphoff, S., Müller, C., Rolinski, S., Waha, K., & Zaehle, S. (2018). Implementing the nitrogen cycle into the dynamic global vegetation, hydrology and crop growth model LPJmL (version 5). *Geoscientific Model Development*, *11*, 2789–2812.

Von Caemmerer, S. (2000). Biochemical models of leaf photosynthesis. PO Box 1139 (150 Oxford Street) Collingwood VIC 3066 Australia: Csiro publishing.

Waha, K., Van Bussel, L., Müller, C, & Bondeau, A. (2012). Climate-driven simulation of global crop sowing dates. Global Ecology and Biogeography, 21(2), 247–259.

Wang, E., Martre, P., Zhao, Z., Ewert, F., Maiorano, A., Rötter, R. P., et al. (2017). The uncertainty of crop yield projections is reduced by improved temperature response functions. *Nature plants*, *3*(8), 17102.

Webber, H., Martre, P., Asseng, S., Kimball, B., White, J., Ottman, M., et al. (2017). Canopy temperature for simulation of heat stress in irrigated wheat in a semi-arid environment: A multi-model comparison. *Field Crops Research*, 202, 21–35.

Wickham, H. (2009). ggplot2: Elegant graphics for data analysis. New York: Springer-Verlag.

Wickham, H. (2011). The split-apply-combine strategy for data analysis. Journal of Statistical Software, 40(1), 1-29.

Willett, W., Rockström, J., Loken, B., Springmann, M., Lang, T., Vermeulen, S., et al. (2019). Food in the Anthropocene: The EAT–Lancet Vommission on healthy diets from sustainable food systems. *The Lancet*, *393*(10170), 447–492.

Wirsenius, Stefan (2000). Human use of land and organic materials: Modeling the turnover of biomass in the global food system. Göteborg, Sweden: Chalmers University of Technology.

Yamori, W., Hikosaka, K., & Way, D. A. (2014). Temperature response of photosynthesis in C<sub>3</sub>, C<sub>4</sub>, and CAM plants: Temperature acclimation and temperature adaptation. *Photosynthesis research*, 119(1-2), 101–117.

Zhao, C., Liu, B., Piao, S., Wang, X., Lobell, D. B., Huang, Y., et al. (2017). Temperature increase reduces global yields of major crops in four independent estimates. *Proceedings of the National Academy of Sciences*, 114(35), 9326–9331.

Zhu, P., Zhuang, Q., Archontoulis, S. V., Bernacchi, C., & Müller, C. (2019). Dissecting the nonlinear response of maize yield to high temperature stress with model-data integration. *Global change biology*, 25(7), 2470–2484.